Author’s response to reviews

Title: The EQ-5D (Euroqol) Is A Valid Generic Instrument For Measuring Quality Of Life In Patients With Dyspepsia

Authors:

Sanjiv Mahadeva (sanjivm1993@yahoo.com)
Hwee-Lin Wee (phawhl@nus.edu.sg)
Khean-Lee Goh (klgoh56@tm.net.my)
Julian Thumboo (julian.thumboo@sgh.com.sg)

Version: 2 Date: 16 February 2009

Author’s response to reviews:

Dear Editor

On behalf of the authors, I would like to thank the reviewers for their comments on the manuscript. They have not asked for any changes, but the journal has requested the name of the ethics committee that approved the study, and this has been added to the "Methods" section of the manuscript.

No additional changes have been made to the manuscript.

Sincerely

Dr S Mahadeva